Author's response to reviews

Title: Acute Medical Unit Comprehensive Geriatric Assessment Intervention Study (AMIGOS): study protocol for a randomised controlled trial

Authors:

Judi A Edmans (judi.edmans@nottingham.ac.uk)
Simon P Conroy (spc3@le.ac.uk)
Rowan Harwood (rowan.harwood@nuh.nhs.uk)
Sarah Lewis (sarah.lewis@nottingham.ac.uk)
Rachel A Elliott (rachel.elliott@nottingham.ac.uk)
Philippa Logan (pip.logan@nottingham.ac.uk)
Lucy Bradshaw (lucy.bradshaw@nottingham.ac.uk)
Matthew Franklin (matthew.franklin@nottingham.ac.uk)
John RF Gladman (john.gladman@nottingham.ac.uk)

Version: 4 Date: 17 August 2011

Author's response to reviews: see over
Dear Sir/Madam,

Thank you for your email requesting the following changes to the format of our manuscript reference MS: 4083150595728918.

1. Authors' contributions: I have reformatted this section as requested.

2. References appearing as 'Error! Reference source not found': I am not sure what happened when this manuscript was uploaded as neither of cases where this appeared were references. Both should say ‘Table 1’ which I have rectified.

3. Figures titles and legends: I have amended these in the manuscript.

4. Table 1 cited in the text: This relates to point 2 above and has now been rectified.

5. Typography: I have checked the typography of the manuscript.

Please let me know if you require any further amendments.

Yours sincerely,

Dr Judi Edmans
Senior Research Fellow
Division of Rehabilitation and Ageing
University of Nottingham